Author's response to reviews

Title: Anti-hormonal Therapy in the Treatment of Ductal Carcinoma of the Parotid Gland

Authors:

Saul Campos-Gomez (saulcampos@gmail.com)
Jose H Flores-Arredondo (josehfloresa@gmail.com)
Rita Dorantes-Heredia (ritadorantesh@yahoo.com.mx)
Mónica Chapa-Ibargüengoitia (monicachapaibar@gmail.com)
Roberto De la Peña-López (rdelap70@yahoo.com)

Version: 2 Date: 4 February 2014

Author's response to reviews:

February 3, 2014

MS: 3876856481138769

Case report

Anti-hormonal Therapy in the Treatment of Ductal Carcinoma of the Parotid Gland

Saul Campos-Gomez, Jose H Flores-Arredondo, Rita Dorantes-Heredia, Mónica Chapa-Ibargüengoitia and Roberto De la Peña-Lopez

BMC Cancer

Ms. Patricia Marie Ratin
Journal Editorial Office
BioMed Central
E: editorial@biomedcentral.com
W: www.biomedcentral.com

Thank you for your review of our manuscript. We have read your comments. Below, I would like to outline changes made.

1) We changed the correct form sections for case reports

Abstract sections: Background, Case presentation and Conclusions
Article sections: Background, Case Presentation, Conclusions, Consent and List of abbreviations used

2.) We added the email addresses of all authors on the Title Page.

3.) We included a Consent Section after conclusions.

4.) We reviewed the guidelines for authors contributions according to ICMJE and reconsidered the authors and keep only whom justify authorship. We used initials to refer to each author’s contribution. We added in acknowledgements section contributor who do not meet the criteria for authorship.

5.) We added a section of no competing Interests

Thanks again for your interest in our work. We await your review of our revised manuscript.

Please do not hesitate to contact us if you have any questions.

With best regards,

Saul Campos-Gomez, MD
Associate Professor, Medical Oncology
Centro Oncológico Estatal, ISSEMYM
Av. Solidaridad Las Torres No. 101
Col. Del Parque
Toluca, México CP 50180